Author’s response to reviews

Title: Consumer Knowledge and Availability Of Maternal And Child Health Services: A Challenge For Achieving MDG 4 And 5 In Southeast Nigeria

Authors:

Nwala K Emmanuel Mr (emmanuel.nwala@yahoo.com)
Ebunoha N Gladys Mrs (gladysebunoha@yahoo.com)
Ugwu U Cosmas Mr (uchecos@yahoo.com)

Version: 5 Date: 24 November 2012

Author’s response to reviews:

Dear Editorial Team,

With reference to the revisions requested before the paper can be published, I hereby submit a corrected draft of the paper. I uploaded two manuscripts: one is a fully formatted draft without track changes, while the other contains track changes that shows the details of the reviews done. The correction covers the requested sections. First, the abstract has been improved with help from two fluent English speaking colleagues. Secondly, the table are well formatted now, thus there are only three tables instead of ten. Thirdly, the fluent English speaking colleagues read through the paper and made reasonable corrections in the language, and this has greatly improved the paper.

However, one author is removed now because she is late.

I strongly believe the paper has fully addressed all aspects of the review requested by the editorial team, and thus should be published.

Many thanks once again.
